Author's response to reviews

Title: Towards a more efficient diabetes control in primary care: six-monthly monitoring compared with three-monthly monitoring in type 2 diabetes - The EFFIMODI trial Design of a randomised controlled patient-preference equivalence trial in primary care

Authors:

Paulien R Wermeling (P.Wermeling@umcutrecht.nl)
Maureen van den Donk (M.vandenDonk@umcutrecht.nl)
Kees J Gorter (K.J.Gorter@umcutrecht.nl)
G. Ardine de Wit (G.A.deWit@umcutrecht.nl)
Yolanda van der Graaf (Y.vanderGraaf@umcutrecht.nl)
Guy EHM Rutten (G.E.H.M.Rutten@umcutrecht.nl)

Version: 4 Date: 7 May 2010

Author's response to reviews:

- In the beginning of the "study design" section I added three sentences
- At the end of the "recruitment of practices and patients" section I changed two sentences